Author’s response to reviews

Title: Thin-section computed tomography-determined usual interstitial pneumonia pattern affects the decision-making process for resection in newly diagnosed lung cancer patients: a retrospective study

Authors:

Naozumi Hashimoto (hashinao@med.nagoya-u.ac.jp)
Akira Ando (andu2727@med.nagoya-u.ac.jp)
Shingo Iwano (iwano45@med.nagoya-u.ac.jp)
Koji Sakamoto (sakakoji@med.nagoya-u.ac.jp)
Shotaro Okachi (s.okachi@med.nagoya-u.ac.jp)
Asuka Matsuzaki (masuka@med.nagoya-u.ac.jp)
Yu Okada (yokada@med.nagoya-u.ac.jp)
Kenji Wakai (wakai@med.nagoya-u.ac.jp)
Kohei Yokoi (k-yokoi@med.nagoya-u.ac.jp)
Yoshinori Hasegawa (yhasega@med.nagoya-u.ac.jp)

Version: 2 Date: 20 Dec 2017

Author’s response to reviews:

To Editor.

We really appreciate your comments for our manuscript.

We revised our manuscript as you kindly suggested.

#1. 1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Response: We added the email addresses for all authors on the title page.

#2. 2. Please amend the headings of the abstract (Background; Methods; Results; Conclusions) and/or of the manuscript (Background; Methods; Results; Discussion; Conclusions) in accordance to BMC Pulmonary Medicine formatting.
Response: We revised the headings of the manuscript (Background; Methods; Results; Discussion; Conclusions) in accordance to BMC Pulmonary Medicine formatting.

#3. 3. Please amend your sentence in 'Population' section: 'This study was also conducted with consent to participate under the 'Ethics, consent and permissions' as it is not very clear.
Response: We deleted this sentence.

#4. 4. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References. Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding, Authors' contributions, and Acknowledgements.
Response: We added the related sentences with the heading 'Declarations' after the Conclusions section and before the References.

#5. 5. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.
Response: We included the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number and approval date in the 'Declarations' section.

#6. 6. Please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.
Response: We included the related statement in “Ethics approval and consent to participate” in the 'Declarations' section.

#7. 7. We note that you have not included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.
Response: We added “Not applicable” for Consent for publication.

#8. 8. Please move the statement regarding the funding from Acknowledgments to Funding section. You may wish to write 'not applicable' in the Acknowledgments.
Response: We moved the statement regarding the funding from Acknowledgments to Funding section. We also added 'not applicable' in the Acknowledgments.
#9. 9. Please amend the initials MA to AM in the Authors' contributions.

Response: We amended the initials MA to AM in the Authors' contributions.

#10. 10. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response: We added the related sentences in the Authors' contributions section;

“All authors read and approved the final version of the manuscript. All authors agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.”

#11. 11. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We uploaded our manuscript as suggested.

#12. 12. Please remove the cover letter from the file inventory, as it is no longer necessary at this stage.

Response: We removed the cover letter.

#13. 13. Please remove the Authorship Change Form from the file inventory, and send it to our Editorial Office separately, so that it can be archived.

Response: We removed the Authorship Change Form. We had already sent the Authorship Change Form to you.